Massachvsetts Institvte of Technology Department of Electrical Engineering and Computer Science

6.313 Contemporary Computer Design

Term Paper Assignment #5 Fall, 1998, Due 9 December

Objective To learn something beyond the topics discussed in class this term. You can expand upon something we have already discussed, or learn about something completely new. This is not intended to be a research project. You do not have to come up with any original work. You should learn about an interesting topic in computer design and describe it in your paper.

Proposal: Due Wednesday 18 November One to two pages typed describing your topic. Please include the bibliographic information for at least two major sources of information for this topic.

Topics A list of possible topics will be available on the course web page:

www.ai.mit.edu/courses/6.313/

Requirements The expected length of the paper is 10-15 pages. The bulk of your sources should be from conferences and journal papers. Textbooks can be used for reference information.

Cite all sources of information. If it is not your own original work, it needs a reference.